Author’s response to reviews

Title: Differentials in the prevalence of anemia among non-pregnant, ever-married women in Bangladesh: Multilevel logistic regression analysis of data from the 2011 BDHS

Authors:

Md Kamruzzaman (kzaman_stat@yahoo.com)
Md Golam Rabbani (mgrpstat57@gmail.com)
Aik Saw (sawaik@hot.com)
Md Abu Sayem (sayem072003@yahoo.com)
Md Golam Hossain (hossain95@yahoo.com)

Version: 5 Date: 1 June 2015

Author’s response to reviews: see over
To
Editor
BMC Women’s Health

Dear Editor

Good morning. We followed your suggestion, we sent our manuscript “MS: 1785788281151373 - Prevalence and associated factors of anemia among non-pregnant women of reproductive age in Bangladesh: Multilevel logistic regression analysis” to Edanz for English correction, and now we have got final corrected manuscript from Edanz. They have changed our manuscript title, new title is “Differentials in the prevalence of anemia among non-pregnant, ever-married women in Bangladesh: Multilevel logistic regression analysis of data from the 2011 BDHS”. I am sending now our corrected manuscript to your Journal. Thank you very much for your cooperation.

With best regards
Dr. Md. Golam Hossain
Professor of Health Research Group
Department of Statistics, University of Rajshahi
Rajshahi-6205, Bangladesh